Author's response to reviews

Title: Knowledge, attitudes, and preventive practices about colorectal cancer among adults in an area of Southern Italy

Authors:

Alessandra Sessa (ale.sessa@hotmail.it)
Rossella Abbate (rossella.abbate@libero.it)
Gabriella di Giuseppe (gabrydg@libero.it)
Paolo Marinelli (paolo.marinelli@unina2.it)
Italo F Angelillo (italof.angelillo@unina2.it)

Version: 4 Date: 9 June 2008

Author's response to reviews:

As requested, each email address is on a new line and it is preceded by the author's initials. We have included details about the previous presentation of the work in the Acknowledgements and it has been removed from the title page.